Author's response to reviews

Title: Comparison of the Epidemiology and Co-morbidities of Heart Failure in the Pediatric and Adult Populations: A Retrospective, Cross-Sectional Study

Authors:

  Gregory Webster (gregory_webster@yahoo.com)
  Jie Zhang (jiezhang00@yahoo.com)
  David N Rosenthal (davidnr@stanford.edu)

Version: 2  Date: 4 April 2006

Author's response to reviews:

We have made the corrections requested in the Minor Essential Revisions, which consisted of removing duplications in the reference section. There were no Major Compulsory Revisions. We appreciate the suggestions in the Discretionary Revisions, but decline to make the change proposed.